Author's response to reviews

Title: Reduced genetic influence on childhood obesity in small for gestational age children

Authors:

Dug Yeo Han (dy.han@auckland.ac.nz)
Angharad R Morgan (ang.morgan@auckland.ac.nz)
Wen J Lam (w.lam@auckland.ac.nz)
John M.D Thompson (j.thompson@auckland.ac.nz)
Rinki Murphy (r.murphy@auckland.ac.nz)
Clare R Wall (c.wall@auckland.ac.nz)
Karen E Waldie (k.waldie@auckland.ac.nz)
Edwin A Mitchell (e.mitchell@auckland.ac.nz)
Lynnette R Ferguson (l.ferguson@auckland.ac.nz)

Version: 2 Date: 14 June 2012

Author's response to reviews: see over
15 June 2012

Dear Editor,

BMC Medical Genetics

We wish to submit our revised manuscript entitled ‘Reduced genetic influence on childhood obesity in small for gestational age children’ to be considered for publication in BMC Medical Genetics.

I have added a paragraph at the end of the Subjects section under Methods regarding the ethics approval numbers and signed contents for study by parents and children.

“The study received ethical approval from the Northern Regional Ethics Committee. The 3.5 year old children’s study had the ethic approval number of 99/097; the study of 7 year old children is AKX/02/00/319; and the study of 11 year old children is NYT/06/00/112. Signed consent for the study and extraction of DNA was given by the parents of the children and also given by the child.”

Kind regards,

Dug Yeo Han